Author's response to reviews

Title: An Intra-tendonous ganglion cyst causing impingement between the anterior cruciate ligament and anterior root of the medial meniscus: a case report.

Authors:

mei guolong (meiguolong1@yahoo.cn)

Version: 6 Date: 25 July 2013

Author's response to reviews: see over
Dear Editor,

Tanks for you attention, I think I have Modified according to your instructions as follow

1. We have removed the images from the main manuscript.
2. Have replaced the title “Case report” with the “Case presentation”
3. replaced the title “Introduction” with the “Background”
4. as you say :we need more references in our article, and We have did the corresponding modification. Such as in Introduction and Discussion.
5. we modified the Conclusion.

With thanks for your consideration,

Mei guo-long